Author’s response to reviews

Title: Perioperative management of paraganglioma and catecholamine-induced cardiomyopathy in child- a case report and review of the literature

Authors:

Xixi Jia (13811452657@163.com)

Xiangyang Guo (xiangyangguoanesth@126.com)

Qing Zheng (cici-user@sohu.com)

Version: 1 Date: 30 Aug 2017

Author’s response to reviews:

To honorable reviewer Brian Donahue:

1. Details of patient management including typical chemistries, coagulation studies, and BNP have been added, and I also made a table to showing changes in vital signs. "the patient was kept intubated for 5 days" has been deleted.

2. Page 9, first paragraph: The sentence "Perioperative mortality associated with Pheochromocytoma resection…" has been deleted.

3. References have been reorganized, including the references suggested.

4. Editing: statements and some nonstandard English usage have been revised.

To honorable reviewer Thomas M. Austin: This article focuses on the anesthetic management of the resection of paraganglioma, previous case reports have been suggested can be used for references, but this article describes the anesthetic management and the changes of vital sign more detailed, and tried to discuss the use of FloTrac/VigileoTM monitoring system in this patient in this revision.